Author's response to reviews

Title: Variability in research ethics review of cluster randomized trials: Scenario-based survey in three countries

Authors:

Monica Taljaard (mtaljaard@ohri.ca)
Jamie C Brehaut (jbrehaut@ohri.ca)
Charles Weijer (cweijer@uwo.ca)
Robert Boruch (robertb@gse.upenn.edu)
Allan Donner (allan.donner@schulich.uwo.ca)
Martin P Eccles (martin.eccles@ncl.ac.uk)
Andrew D McRae (Andrew.McRae@albertahealthservices.ca)
Raphael Saginur (rsaginur@ohri.ca)
Merrick Zwarenstein (merrick.zwarenstein@ices.on.ca)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)

Version: 3
Date: 20 January 2014

Author's response to reviews:

In addition to the revisions to our manuscript, the Editorial Assistant requested a revision to the Author Contributions section to mention each author individually. This has been done.